**Author's response to reviews**

**Title:** MOSAIC (MOthers' Advocates In the Community): protocol and sample description of a cluster randomised trial of mentor mother support to reduce intimate partner violence among pregnant or recent mothers.

**Authors:**

Angela J Taft (a.taft@latrobe.edu.au)
Rhonda Small (r.small@latrobe.edu.au)
Kelsey L Hegarty (k.hegarty@unimelb.edu.au)
Judith Lumley (j.lumley@latrobe.edu.au)
Lyndsey F Watson (l.watson@latrobe.edu.au)
Lisa Gold (lisa.gold@deakin.edu.au)

**Version:** 5  **Date:** 22 May 2009

**Author's response to reviews:**

Abstract - PSI-SF (correct hyphen), removeal of 'the' prior to 'MOSAIC's impact para 2.
P.7 -addition of 'was' to acceptable dot point 5.
P.9. 2nd para -removal of 'required'.
P. 20. Authors' information - correct spacing and additon of 'former Director' to Prof Lumley's citation.
Tables 1, 2. Right formattting of n and % to each table for correct alignment.